Author’s response to reviews

Title: 118 SNPs of Folate-Related Genes and Risks of Selected Congenital Anomalies

Authors:

Gary M Shaw (GShaw@marchofdimes.com)
Wei Lu (wlu@ibt.tamhsc.edu)
Huiping Zhu (hzhu@ibt.tamsch.edu)
Wei Yang (WY@marchofdimes.com)
Farren BS Briggs (fbriggs@genepi.berkeley.edu)
Suzan L Carmichael (SCarmichael@marchofdimes.com)
Lisa F Barcellos (barcello@genepi.berkeley.edu)
Edward J Lammer (elammer@chori.org)
Richard H Finnell (rfinnell@ibt.tamhsc.edu)

Version: 5 Date: 8 May 2009

Author’s response to reviews: see over
May 8, 2009

BMC Genomics
Publications Department

Dear Editor,

Please find attached our revised manuscript, “118 SNPs of Folate-Related Genes and Risks of Spina Bifida and Conotruncal Heart Defects”. We have specifically revised the requested areas. Those are: 1) we have formatted the abstract as requested; 2) we have included a sentence in the Methods about informed consent regarding the use of bloodspots; 3) we have included a "Competing Interests” section; and 4) we have included an "Authors Contributions” section.

We look forward to hearing from you soon about the final acceptance of this manuscript.

Sincerely,

Gary M. Shaw, DrPH
Professor
Department of Pediatrics
Division of Neonatal & Developmental Medicine
Stanford University
Medical School Office Bldg
251 Campus Dr. Room x159
Stanford, CA 94305-5460

Telephone: (650) 721-5746
Fax: (650) 721-5751
Email: gmshaw@stanford.edu